# WASHINGTON STATE BOARD FOR COMMUNITY AND TECHNICAL COLLEGES 401(a) RETIREMENT PLAN Restatement Effective Date: January 1, 2016

# TABLE OF CONTENTS

DEF	TNITIONS	•••••
1.1	"Anniversary Date"	
1.2	"Appointing Authority"	
1.3	"Benefit Administrator"	
1.4	"Break In Service"	
1.5	"Code"	
1.6	"Compensation"	
1.7	"Contract"	
1.8	"Deferred Compensation"	
1.9	"Designated Beneficiary"	
1.10	"Eligible Employee"	
1.11	"Employee"	
1.12	"Employee Contribution"	
1.13		
1.14	"Fiscal Year"	
	"Former Participant"	
	"415 Compensation"	
	"Investment Advisory Committee"	
	"Investment Manager"	
1.19	"Late Retirement Date"	
1.20		
1.21		
1.22	"Participant's Account"	
1.23	"Participant's Employer Contributions Account"	
1.24	"Participant's Employee Contributions Account"	
	"Participant's Rollover Account"	
	"Participating Employer"	
1.27	"Pick-up Contributions"	
1.28	"Plan"	
1.29	"Plan Administrator"	
1.30	"Plan Fund"	
1.31	"Plan Year"	
1.32	"Predecessor Plan"	
1.33	"Pre-Restatement Plan"	
1.34	"RCW"	
1.35	"Restatement Effective Date"	
1.36	"Regulation"	
1.37	"Retired Participant"	
1.38	"Retirement Date"	
1.39	"Sponsor"	
1.40	"State Board"	
1.41	1	
1.42	======	
1.43	"Valuation Date"	
1.44	"Vested"	
1.45	"Year Of Service"	
_	LAN SPONSOR	

	2.2	Authority To Appoint Benefit Administrator	10
	2.3	Allocation And Delegation Of Responsibilities	11
	2.4	Appointment Of Advisers	11
	2.5	Information From Participating Employers	11
3.	E	LIGIBILITY	1
	3.1	Conditions Of Eligibility	11
	3.2	Salary Reduction Agreement	13
	3.3	Effective Date Of Participation	13
	3.4	Determination Of Eligibility	13
	3.5	Omission Of Eligible Employee	14
	3.6	Inclusion Of Ineligible Employee	14
	3.7	Reemployment Of Former Participant	14
	3.8	Election Not To Participate	14
4.	C	ONTRIBUTIONS AND ALLOCATIONS	<b>1</b> 4
	4.1	Employee Contributions	14
	4.2	Employer Contributions	15
	4.3	Allocation Of Contributions and Earnings	15
	4.4	Maximum Annual Additions Under Code Section 415	16
	4.5	Transfers And Rollovers From Eligible Retirement Plans	17
5.	A	CCOUNTS, INVESTMENTS AND VALUATIONS	19
	5.1	Individual Accounts	19
	5.2	Allocation Of Earnings And Losses To Accounts	19
	5.3	Account Statements	19
	5.4	Permissible Investments	20
	5.5	Purpose Of The Plan Fund	20
	5.6	Exclusive Benefit Of Participants And Beneficiaries	20
	5.7	Benefits Supported Only By The Plan Fund	20
	5.8	Payment Of Expenses	21
6.	D	ETERMINATION AND DISTRIBUTION OF BENEFITS	21
	6.1	Determination Of Benefits Upon Retirement	21
	6.2	Determination Of Benefits Upon Termination	22
	6.3	Distribution Of Benefits; Spousal Consent	23
	6.4	Distribution Of Benefits Upon Death	25
	6.5	Distribution For Minor Beneficiary	27

6.6	Location Of Participant Or Beneficiary Unknown	27
6.7	Determination Of Benefits In Event Of Disability	27
6.8	Advance Distribution For Hardship	28
6.9	Qualified Domestic Relations Orders	29
6.10	0 Direct Rollover	29
<b>7.</b> ]	PLAN ADMINISTRATRATION	30
7.1	Plan Administrator	30
7.2	2 Investment Advisory Committee; State Board Powers	31
7.3	Indemnity, Bonding And Insurance	32
7.4	Fiduciaries	32
7.5	Claims And Appeals Procedures	33
8.	AMENDMENT, TERMINATION AND MERGERS	35
8.1	Amendment	35
8.2	2 Termination	36
8.3	Merger Or Consolidation	36
<b>9.</b> 1	MISCELLANEOUS	36
9.1	Participant's Rights	36
9.2	2 Alienation	36
9.3	Construction Of Plan	37
9.4	Gender And Number	37
9.5	Recovery Of Erroneous Contributions	37
9.6	Sponsor's Protective Clause	37
9.7	Receipt And Release For Payments	38
9.8	Action By An Employer	38
9.9	Headings	38
9.10	0 Uniformity	38
9.1	1 Compliance With Userra And Related Laws	38
9.12	2 Prohibition Against Diversion Of Funds	38
<b>10.</b> ]	PARTICIPATING EMPLOYERS	39
10.	.1 Requirements Of Participating Employers	39
10.2	.2 Designation Of Agent	39
10.	.3 Participating Employer's Contribution	39
10.4	.4 Amendment	40
10.:	.5 Plan Administrator's Authority	40

11. MINIMUM DISTRIBUTION REQUIREMENTS	40
11.1 General Rules	40
11.2 Time And Manner Of Distribution	40
11.3 Required Minimum Distributions	42
11.4 Required Minimum Distributions After Death	42
11.5 Definitions	43
Appendix A: List of Participating Employers	46

# WASHINGTON STATE BOARD FOR COMMUNITY AND TECHNICAL COLLEGES 401(a) RETIREMENT PLAN

#### INTRODUCTION

WHEREAS, effective January 1, 2011, Washington State Board for Community and Technical Colleges (herein referred to as the "Sponsor"), a governmental entity, adopted a restatement of the Washington State Board for Community Colleges 401(a) Retirement Plan (the "Plan") in order to recognize the contribution made by the Eligible Employees of Participating Employers and to reward such contribution by means of Plan benefits for those Eligible Employees who qualify as Participants hereunder; to take advantage of certain tax-favored contribution options that allow certain Employee Contributions (sometimes called "Pick-Up Contributions") to be treated as Employer Contributions for tax purposes under Code Section 414(h); and to provide certain qualifying Participants with a supplemental annual retirement income as specified in separate retirement plan provisions under state law (and referred to herein as a Supplemental Retirement Benefit), exclusive of Federal Old Age Survivors Insurance Benefits; and

WHEREAS, the Sponsor is a governmental entity not subject to the provisions of the Employee Retirement Income Security Act of 1974 ("ERISA") nor subject to certain coverage and non-discrimination provisions of the Code, and whereas the Sponsor does not intend to voluntarily subject the Plan to any other provisions of ERISA or the Code that are not required of a governmental plan; and

WHEREAS, the Plan received a favorable determination letter from the Internal Revenue Service that did not cover the supplemental annual retirement income benefit; and

WHEREAS, the Sponsor wishes to provide the supplemental benefits in a separate retirement plan that is intended to be qualified under Code Section 401(a), et seq., and to make certain other changes.

NOW, THEREFORE, effective January 1, 2016, the Sponsor hereby adopts this restated Plan, which is intended to function as a Code Section 401(a) qualified Plan, and a related trust, for the purpose of further fulfilling the purposes stated above.

#### 1. **DEFINITIONS**

1.1 "Anniversary Date"

"Anniversary Date" means December 31.

# 1.2 "Appointing Authority"

"Appointing Authority" means a Participating Employer's governing board or the designees of such boards.

#### 1.3 "Benefit Administrator"

"Benefit Administrator" means the service provider to the Plan that is responsible for maintaining the records of the Participants' Accounts and determining benefits under the Plan, and for performing certain other benefit administration and communications functions, in accordance with such contractual agreements and direction letters as may exist from time to time between such service provider and the Plan Administrator or Sponsor. The Benefit Administrator is sometimes referred to as the "Record Keeper" in certain documents such as the trust agreement with the Trustee. As of the Restatement Effective Date, the Benefit Administrator is Teachers Insurance and Annuity Association of America.

#### 1.4 "Break in Service"

"Break in Service" means termination of all employment with a Participating Employer for a full academic year quarter or an equivalent period of time, excluding summer quarter or an equivalent off-season quarter.

#### 1.5 "Code"

"Code" means the Internal Revenue Code of 1986, as amended from time to time, and regulations issued pursuant thereto.

#### 1.6 "Compensation"

"Compensation" with respect to any Participant means the sum of the following elements of compensation paid to the Participant from his or her Participating Employer, including base salary, summer quarter compensation, extra duty pay, leave stipends, and grants made by or through the Participating Employer or State Board; but not including any severance pay, early retirement incentive payment, remuneration for unused sick or personal leaves, settlement payments resulting from claims, disputes or litigation, or remuneration for unused annual or vacation leaves in excess of the amount payable for thirty days or two hundred forty hours of service. For purposes of this Section, the determination of Compensation shall be made by including both: (i) amounts which are contributed by the Participating Employer pursuant to a salary reduction agreement and which are not includable in the gross income of the Participant under Code Sections 125, 132(f)(4), 402(e)(3), 402(h), 403(b) or 457, and (ii) Employee Contributions (sometimes called Pick-Up Contributions).

For a Participant's initial year of participation, Compensation shall be recognized as of such Employee's effective date of participation pursuant to Section 3.3.

Compensation in excess of the applicable dollar limitation under Code Section 401(a)(17) that is in effect for the applicable Plan Year (e.g., \$210,000 for 2005) shall be disregarded. Such dollar limitation shall be adjusted as determined by the Secretary of the Treasury in accordance with Code Section 401(a)(17), except that the dollar increase in effect on January 1 of any calendar year shall be effective for the Plan Year beginning with or within such calendar year. If Compensation for any prior determination period is taken into account in determining an Employee's benefits accruing in a Plan Year beginning after December 31, 1993, the Compensation for that prior determination period is subject to the adjusted compensation limit in effect for that prior determination period. For this purpose, for determination periods beginning before the first day of the first Plan Year beginning on or after January 1, 1994, the annual compensation limit is \$150,000. For any short Plan Year the Compensation limit shall be an amount equal to the Compensation limit for the calendar year in which the Plan Year begins multiplied by the ratio obtained by dividing the number of full months in the short Plan Year by twelve (12).

For purposes of this Section 1.6, amounts under Code Section 125 include any amounts not available to a Participant in cash in lieu of group health coverage because the Participant is unable to certify that he or she has other health coverage. An amount will be treated as an amount under Code Section 125 only if the Participating Employer does not request or collect information regarding the Participant's other health coverage as part of the enrollment process for the health plan.

#### 1.7 "Contract"

"Contract" means any annuity contract (group or individual), issued pursuant to the terms of the Plan. The term Contract includes, by way of example, group retirement annuity contracts ("GRA" contracts) and retirement annuity contracts ("RA" contracts, transferred to this Plan from the Predecessor Plan) issued by Teachers Insurance and Annuity Association ("TIAA") and the College Retirement Equities Fund ("CREF"), and investments in any such Contract shall be governed by the terms of such Contract.

## 1.8 "Deferred Compensation"

"Deferred Compensation" with respect to any Participant means the aggregate amount of the Participant's Employee Contributions that have been contributed to this Plan pursuant to Section 4.1, excluding any such amounts distributed as "excess additions" pursuant to Section 4.6.

#### 1.9 "Designated Beneficiary"

"Designated Beneficiary" means the person to whom the remaining balance of a deceased Participant's total account is payable. The Designated Beneficiary shall be the spouse (if any) to whom the Participant was married as of his or her benefit commencement date (or date of death, if earlier), or, with the consent of such spouse, if any, such other person or

persons as shall have been nominated by written designation duly executed and filed with either the Benefit Administrator, the retiree's Participating Employer, or the State Board.

# 1.10 "Eligible Employee"

"Eligible Employee" means any Employee who meets the conditions for eligibility in Section 3.1. Provided, however, that the definition of Eligible Employee shall be determined based on the payroll records of the respective Participating Employer, and shall not include any "leased employee" (as defined by Code Section 414(n)) or any person who, on the payment records of the Participating Employer, is treated as an independent contractor or as an employee of a third-party temporary services provider.

# 1.11 "Employee"

"Employee" means any person who is a common law employee of a Participating Employer, but excludes any person who performs services as an independent contractor. The definition of Employee includes any "leased employee" (as defined by Code Section 414(n)) of a Participating Employer.

#### 1.12 "Employee Contribution"

"Employee Contribution" is synonymous with the term "Pick-Up Contribution," and means a contribution to the Plan as described in Section 4.1.

# 1.13 "Employer Contribution"

"Employer Contribution" means any Participating Employer contributions to the Plan that are made pursuant to Section 4.2.

#### 1.14 "Fiscal Year"

"Fiscal Year" means the Sponsor's accounting year of 12 months commencing on July 1st of each year and ending the following June 30th.

# 1.15 "Former Participant"

"Former Participant" means a person who has been a Participant, but who has ceased to be a Participant for any reason.

#### 1.16 "415 Compensation"

"415 Compensation" with respect to any Participant means the sum of:

(i) such Participant's wages as defined for Federal Income Tax purposes in Code Section 3401(a) and all other payments of compensation by the Participating Employer (in the course of the Participating Employer's trade or business) for a Plan Year for which the Participating Employer is required to furnish the Participant a written statement on IRS Form W-2 under Code Sections 6041(d), 6051(a)(3) and 6052; and, for Plan Years

commencing on or after January 1, 1998, "415 Compensation" shall also include the following:

- (ii) such Participant's pre-tax contributions to a cafeteria plan under Code Sections 125 and to a qualifying commuting or parking program under Code Section 132(f)(4); and
- (iii) such Participant's elective deferrals under Code Section 402(e)(3), 402(h), 403(b) or 457.

"415 Compensation" must be determined without regard to any rules under Code Section 3401(a) that limit the remuneration included in wages based on the nature or location of the employment or the services performed (such as the exception for agricultural labor in Code Section 3401(a)(2)).

For purposes of this Section 1.16, amounts under Code Section 125 include any amounts not available to a Participant in cash in lieu of group health coverage because the Participant is unable to certify that he or she has other health coverage. An amount will be treated as an amount under Code Section 125 only if the Participating Employer does not request or collect information regarding the Participant's other health coverage as part of the enrollment process for the health plan.

For limitation years beginning on or after January 1, 2008, "415 Compensation" shall include amounts paid after severance from employment with a participating Employer to the extent permitted by Treasury Regulations Section 1.415(c)-2(e)(3) and shall exclude all other amounts paid after severance from employment with a participating Employer as required by Treasury Regulations Section 1.415(c)-2(e)(3)(iv).

Effective January 1, 2009, "415 Compensation" shall include any payment which is:

- (i) Made by a Participating Employer to individuals with respect to any period during which they are performing service in the uniformed services while on active duty for a period of more than 30 days; and
- (ii) Represents all or part of the wages that such individuals would have received from the Participating Employer if such individuals were performing services for the Participating Employer.
- 1.17 "Investment Advisory Committee"

"Investment Advisory Committee" means a committee consisting of a chairperson, plus at least two additional voting members appointed by the chairperson. Except to the extent determined otherwise by the State Board, the Deputy Executive Director of Business Operations for the State Board shall serve as the chairperson of the Investment Advisory Committee. Effective on and after January 1, 2006, the Investment Advisory Committee

shall have the authority and responsibility to monitor the performance of any of the Contracts or other investment fund options of the Plan Fund and of any Investment Manager, and to provide advice to the State Board and Plan Administrator on the selection, retention, elimination or replacement at any time any and all of the Contracts and other investment fund options available for Participant investment direction under this Plan, and/or advice on the appointment, discharge or replacement of any Investment Manager.

# 1.18 "Investment Manager"

"Investment Manager" means an entity that (a) has the power to manage, acquire, or dispose of Plan assets and (b) acknowledges fiduciary responsibility to the Plan in writing. Such entity must be a person, firm, or corporation registered as an investment adviser under the Investment Advisers Act of 1940, a bank, or an insurance company.

#### 1.19 "Late Retirement Date"

"Late Retirement Date" means the first day of the month coinciding with or next following a Participant's actual Retirement Date after having reached his or her Normal Retirement Date.

#### 1.20 "Normal Retirement Date"

"Normal Retirement Date" means, as applied to a Participant, the earliest to occur of any of the three following dates: (a) his or her attainment of age fifty-five (55), the date of his or her completion of thirty (30) Years of Service, or the date of termination of employment as a result of a disability, where such disability status has been determined by the Participant's Participating Employer after consideration of a medical evaluation.

#### 1.21 "Participant"

"Participant" means any Eligible Employee, Terminated Participant or Retired Participant who meets the conditions for participation in this Plan and has commenced participation in accordance with Section 3 of this Plan, and who has not yet ceased to participate in this Plan as a result of death or receipt of all vested benefits that are due or payable hereunder.

#### 1.22 "Participant's Account"

"Participant's Account" means an account established and maintained by the Benefit Administrator for each Participant with respect to his or her total interest in the Plan including the Participant's Employee Contributions Account, the Participant's Employer Contributions Account and (if any) the Participant's Rollover Account.

#### 1.23 "Participant's Employer Contributions Account"

"Participant's Employer Contributions Account" means an account established and maintained by the Benefit Administrator for each Participant with respect to the Participant's interest in the Plan resulting from Employer Contributions pursuant to Section 4.2.

#### 1.24 "Participant's Employee Contributions Account"

"Participant's Employee Contributions Account" means an account established and maintained by the Benefit Administrator for each Participant with respect to his or her interest in the Plan resulting from mandatory Employee Contributions which have been "picked up" by the Participating Employer pursuant to Section 4.1 and Code Section 414(h)(2).

### 1.25 "Participant's Rollover Account"

"Participant's Rollover Account" means an account established and maintained by the Benefit Administrator for each Participant with respect to his or her interest in the Plan resulting from the transfer to the Plan of funds from another Eligible Retirement Plan pursuant to Section 4.5.

#### 1.26 "Participating Employer"

"Participating Employer" means an educational organization described in Code Section 170(b)(1)(A)(ii) operated by the State of Washington or an agency or instrumentality of the foregoing, which is the common law employer of one or more Eligible Employees or former Eligible Employees, and which is an employing entity designated by the Sponsor to participate in the Plan. The Participating Employers include the State Board, the Washington Student Achievement Council and other educational institutions in the State of Washington listed on Appendix A of this Plan, as it may be amended from time to time.

#### 1.27 "Pick-up Contributions"

"Pick-up Contributions" means Employee Contributions that are treated as employer contributions for determining tax treatment under Code Section 414(h)(2). Pick-up Contributions include payments from Employees on leave of absence due to disability or former Employees who have terminated due to disability and are receiving a salary continuation benefit through a plan made available by the State of Washington.

#### 1.28 "Plan"

"Plan" means this instrument, including all amendments thereto.

#### 1.29 "Plan Administrator"

"Plan Administrator" means the persons or entities designated by the Sponsor pursuant to Article 7 to administer the Plan on behalf of the Sponsor. As of the Restatement Effective Date, the Plan Administrator is the Deputy Executive Director of Business Operations of the State Board.

#### 1.30 "Plan Fund"

"Plan Fund" means the assets of the Plan as the same shall exist from time to time. Prior to the Restatement Effective Date, the Plan Fund was invested exclusively in Contracts

meeting the requirements of Code Section 403(a). On and after the Restatement Effective Date, the Plan Fund shall include a combination of (i) Contracts (which shall be deemed to be held in trust in accordance with Code Section 401(f)), and (ii) other Plan assets, including, by way of example, shares or units of mutual funds or other investment vehicles which shall be held in trust by the Trustee. Notwithstanding any other provision of the Plan, the Plan Fund, including all amounts of Deferred Compensation, all property and rights purchased with those amounts and all income attributable thereto, shall be held solely for the purposes set forth in this Plan.

#### 1.31 "Plan Year"

"Plan Year" means the Plan's accounting year of twelve (12) months commencing on January 1st of each year and ending the following December 31st.

#### 1.32 "Predecessor Plan"

"Predecessor Plan" means the Washington State Board for Community and Technical Colleges 403(b) Plan that was in effect prior to January 1, 1998.

#### 1.33 "Pre-Restatement Plan"

"Pre-Restatement Plan" means the Washington State Board for Community and Technical Colleges 403(a) Retirement Plan, as such plan was in effect from January 1, 1998 to December 31, 2005.

# 1.34 "RCW"

"RCW" means the Revised Code of Washington, as it may be amended from time to time.

#### 1.35 "Restatement Effective Date"

"Restatement Effective Date" means January 1, 2006, that being the effective date of the conversion of the Pre-Restatement Plan from its qualified status under Code Section 403(a) to qualified status under Code Section 401(a). In 2010, the plan was amended to comply with Code and in 2011 to comply with RCW changes effective July 1, 2011.

# 1.36 "Regulation"

"Regulation" means the Income Tax Regulations as promulgated by the Secretary of the Treasury or his delegate, and as amended from time to time.

#### 1.37 "Retired Participant"

"Retired Participant" means a person who has been a Participant, but who has become entitled to Retirement Benefits under the Plan and has not yet ceased to be a Participant as a result of receiving all vested benefits which are due or payable hereunder.

#### 1.38 "Retirement Date"

"Retirement Date" means the date on which a Participant retires (the effective date of commencement of benefits hereunder), whether such retirement occurs on a Participant's Normal Retirement Date or Late Retirement Date. The Retirement Date for Participants opting for a Phased Retirement, as described in Section 6.1, shall mean the same as Late Retirement Date.

#### 1.39 "Sponsor"

"Sponsor" means Washington State Board for Community and Technical Colleges, any successor which shall sponsor this Plan, and any predecessor which has sponsored this Plan.

#### 1.40 "State Board"

"State Board" means the Washington State Board for Community and Technical Colleges, as created in RCW 28B.50.050.

# 1.41 "Terminated Participant"

"Terminated Participant" means a person who has been a Participant, but whose employment has been terminated other than by death, disability or retirement and who has not yet ceased to be a Participant as a result of receiving all Vested benefits which are due or payable hereunder.

#### 1.42 "Trustee"

"Trustee" means JPMorgan Chase Bank, N.A., in its capacity as trustee of certain Plan assets under the terms of a trust agreement with the Sponsor, or any entity which the Sponsor may in its discretion subsequently appoint at any time to hold assets in trust under this Plan. Effective January 1, 2016, "Trustee" means TIAA-CREF Trust Company, FSB.

#### 1.43 "Valuation Date"

"Valuation Date" means, for the investment funds held by the Trustee that are valued daily, each day of the Plan Year that the New York Stock Exchange is open. The Plan Administrator may also designate any additional date as a "special Valuation Date."

#### 1.44 "Vested"

"Vested" means the non-forfeitable portion of any Account maintained on behalf of a Participant. Under the terms of this Plan, the entire balance of a Participant's Account shall be 100% Vested immediately, and at all times.

#### 1.45 "Year of Service"

"Year of Service" means retirement credit based on unbroken full time employment or the equivalent thereof based on part-time employment in an eligible position in a Fiscal Year

during which Plan contributions were made by a Washington public higher education institution to the State Board Retirement Plan or any other plan established pursuant to RCW 28B.10.400. Any year or fractional year of prior service in a Washington Public Retirement System while employed at a Participating Employer is included, provided the Participant transferred directly from such retirement system to the State Board Retirement Plan and will receive a pension benefit from such other retirement system; and provided further, that not more than one Year of Service will be credited for service in any one Fiscal Year. For this purpose, "unbroken service" means service without a Break in Service. Except as otherwise provided for within this Plan Document, periods of leave without pay or other periods in which a Participant is not earning Compensation from a Participating Employer and periods after termination of employment by reason of permanent disability while receiving a salary continuation benefit through a plan made available by the State of Washington shall not be included in a Participant's Years of Service.

#### 2. PLAN SPONSOR

#### 2.1 POWERS AND RESPONSIBILITIES OF THE SPONSOR

- (a) In addition to the authority and responsibilities otherwise provided for in this Plan, the Sponsor has the authority to amend, merge or terminate the Plan, and, to appoint and remove a Plan Administrator. The Sponsor may delegate its authority and duties as delineated in this Plan Document to the Plan Administrator. Effective on and after the Restatement Effective Date, and in the absence of a subsequent Plan amendment to the contrary, the Plan Administrator shall be the Deputy Executive Director of Business Operations of the State Board.
- (b) The Sponsor shall have the authority and responsibility to appoint and replace the Benefit Administrator. The Benefit Administrator shall have such duties as may be stated in a contract between that service provider and either the State Board or the Plan Administrator.
- (c) Effective on and after the Restatement Effective Date, the State Board shall have the authority and responsibility to appoint the Trustee for the purpose of holding any or all Plan assets.

#### 2.2 AUTHORITY TO APPOINT BENEFIT ADMINISTRATOR

The Sponsor, by action of the State Board, may appoint one or more Benefit Administrators. As of the Restatement Effective Date, the Benefit Administrator is Teachers Insurance and Annuity Association of America.

Any person or entity shall be eligible to serve as a Benefit Administrator. In the case of any Benefit Administrator that is a third-party service provider (rather than an Employee of the State Board or a Participating Employer), the contractual terms applicable to the Benefit Administrator shall be stated in a written agreement between the Benefit Administrator and the State Board (or Plan Administrator). A Benefit Administrator may resign by delivering a written resignation,

subject to the terms and any notice requirements specified in the applicable contract. Any Benefit Administrator may be removed by the Sponsor by delivery of written notice of removal, to take effect at a date specified therein, or upon delivery to the Benefit Administrator if no date is specified, subject to the terms and any notice requirements specified in the applicable contract.

The Sponsor, upon the resignation or removal of a Benefit Administrator, shall promptly designate in writing a successor to this position. If the Sponsor does not appoint a successor Benefit Administrator, the Plan Administrator will function as the Benefit Administrator.

#### 2.3 ALLOCATION AND DELEGATION OF RESPONSIBILITIES

If more than one person or entity is appointed as Benefit Administrator, the responsibilities of each Benefit Administrator may be specified by the Sponsor and accepted in writing by each Benefit Administrator, in accordance with the terms of a contractual agreement between the Benefit Administrator and the Sponsor or Plan Administrator.

#### 2.4 APPOINTMENT OF ADVISERS

The Sponsor may appoint counsel, specialists, advisers, and other persons as the Sponsor deems necessary or desirable in connection with its sponsorship and oversight of this Plan.

#### 2.5 INFORMATION FROM PARTICIPATING EMPLOYERS

To enable the Plan Administrator and Benefit Administrator to perform their functions, any Participating Employer shall supply full and timely information to the Plan Administrator and/or Benefit Administrator on all matters relating to the Compensation of all Participants, their hours of service, their Years of Service, their retirement, death or termination of employment, and such other pertinent facts as the Plan Administrator or Benefit Administrator may require. The Plan Administrator and Benefit Administrator may rely upon such information as is supplied by the Participating Employer and shall have no duty or responsibility to verify such information.

#### 3. ELIGIBILITY

#### 3.1 CONDITIONS OF ELIGIBILITY

(a) Excepted as otherwise noted in this section, eligibility in this Plan is limited to Eligible Employees of a Participating Employer who hold appointments as full-time or part-time faculty or administrators/professionals exempt from the provisions of 41.06 RCW and who are assigned a cumulative total of at least fifty percent of a full-time workload as defined by the collective bargaining agreement and/or Appointing Authority at one or more Participating Employers for at least two consecutive college quarters or its equivalent. For faculty, the percent of full time will be calculated consistent with RCW 28B.50.489.

Participation in the plan by employees of the Washington Student Achievement Council is limited to Eligible Employees who have contributed premiums to a similar qualified plan.

Participating Employers shall notify their newly hired Eligible Employees of their potential right to participate. Employees who have established eligibility with a previous employer are responsible for notifying their new college or state board employer of their potential right to maintain that participation. For the purposes of maintaining participation and/or establishing initial eligibility, spring and fall quarters shall be considered as consecutive periods of employment.

Employees of Participating Employers establishing Plan eligibility on or after July 1, 2011 and who are receiving or eligible to receive a retirement allowance under Title 41 RCW or chapter 43.43 RCW are prohibited from participation.

- (b) Participation in the Plan is permitted for current and former Employees of a Participating Employer who are on leave of absence or who have terminated employment by reason of permanent disability who are receiving a salary continuation benefit through a plan made available by the State of Washington.
- Co An Employee who moves from an ineligible to an eligible position for the same Participating Employer may become a Participant by so electing within thirty (30) days following such move. The effective date of participation is the date the election is received by the Participating Employer. In the event that an Eligible Employee makes a timely and valid election to participate in the Plan, the Employee shall be enrolled as a Participant and contributions made as soon as administratively practical following the completion of such election procedure. In the event an Eligible Employee fails to make a timely and valid election to participate in the Plan, the Eligible Employee will be defaulted into the Plan with the effective date of participation being the default date.
- (d) A Participant who moves from an eligible to an ineligible classified position for the same Participating Employer may continue to be a Participant by so electing within thirty (30) days following such move. The effective date of an election to no longer participate in the Plan is the date the election is received by the Participating Employer. Failure to make a timely and valid election will result in continued participation in this Plan.
- (e) Except as otherwise provided in this Section, Participants shall continue participation regardless of the proportion of full-time duties assigned as long as continuously employed in an eligible position by a Participating Employer. For this purpose, "continuously employed" means employment without a Break in Service.
- (f) Employees who become eligible on or after July 1, 2011 may elect to establish membership in Plan 3 of the Washington Public Employees Retirement System or, if faculty, in the Washington State Teachers Retirement System within thirty (30) days of initial eligibility. New Employees who at the time of employment are members of the Washington State Teachers Retirement System or the Washington Public Employees Retirement System may elect to participate in this Plan or may elect to continue to participate in the Washington State Teachers Retirement System or the Washington Public Employees Retirement System so long as they are eligible as provided in the Washington Administrative Code adopted by the Washington State

Department of Retirement Systems. Employees failing to make an election within 30 days will default into the Plan. The elections provided in this section apply as Employees transfer between Participating Employers and are irrevocable as long as the Participant is continuously employed by a Participating Employer.

- (g) State law prohibits members of the Washington Public Employee Retirement System to concurrently participate in more than one state funded plan. Employees meeting Plan eligibility who are concurrently employed with a public employer and actively contributing to the Washington Public Employee Retirement System are required to notify their Participating Employer and be given the following options:
- (1) Participate in the Plan, forgoing active membership (contributions and service credit) in Washington Public Employee Retirement System with their other employer; or
- (2) Continue active participation in the Washington Public Employee Retirement System with the other employer, forgoing participation in the Plan.

Failure to make an election within 30 days of notification will result in default into the Plan. The Participating Employer is required to notify the Department of Retirement Systems of the election to become a Participant. The Participant is required to notify the other employer of his or her election to participate in the Plan. This irrevocable election remains in effect as long as the employee is actively participating in the Public Employee Retirement System with the other employer.

#### 3.2 SALARY REDUCTION AGREEMENT

Each Eligible Employee who elects to participate in the Plan is required to complete an irrevocable salary reduction agreement as a condition of employment.

#### 3.3 EFFECTIVE DATE OF PARTICIPATION

An Eligible Employee meeting the criteria in Section 3.1 shall become a Participant on the date upon which the employee submits his or her election to participate to the Participating Employer or is defaulted into the Plan, provided said Employee was still employed as of such date (or if not employed on such date, as of the date of rehire) in an Eligible Position by a Participating Employer. Contributions will be made as soon as administratively possible after the Eligible Employee submits his or her election and the salary reduction agreement.

#### 3.4 DETERMINATION OF ELIGIBILITY

An Employee's Participating Employer shall determine the eligibility of the Employee for participation in the Plan in accordance with Section 3.1. Such determination shall be conclusive and binding upon all persons, as long as the determination is made in accordance with the terms of the Plan.

# 3.5 OMISSION OF ELIGIBLE EMPLOYEE

If, in any Plan Year, any Employee who should be included as a Participant in the Plan is erroneously omitted and discovery of such omission is not made until after a contribution for the year has been made by his or her Participating Employer, the Participating Employer shall make a subsequent contribution with respect to the omitted Employee in the amount that the said Participating Employer would have contributed with respect to the Participant had he or she not been omitted.

#### 3.6 INCLUSION OF INELIGIBLE EMPLOYEE

If, in any Plan Year, any person who should not have been included as a Participant in the Plan is erroneously included and discovery of such incorrect inclusion is not made until after a contribution for the year has been made, the Employer shall be entitled to recover the contribution made with respect to the ineligible person. In such case, the amount of the erroneous Employer Contribution shall be remitted from the Plan Fund to the Participating Employer, and the Participating Employer shall pay the amount of the erroneous Employee Contribution to the Employee (after withholding any applicable taxes).

#### 3.7 REEMPLOYMENT OF FORMER PARTICIPANT

If any Former Participant is reemployed by a Participating Employer he or she shall resume participation in the Plan subject to reestablishing eligibility in accordance with Section 3.1 in the same manner as if such termination had not occurred.

#### 3.8 ELECTION NOT TO PARTICIPATE

All Eligible Employees shall participate in the Plan unless any such new Employee has the option and elects to participate in Washington State Teachers Retirement System Plan or the Washington Public Employees Retirement System Plan as described in Section 3.1.

#### 4. CONTRIBUTIONS AND ALLOCATIONS

#### 4.1 EMPLOYEE CONTRIBUTIONS

- (a) The Participating Employer shall reduce an Eligible Participant's Compensation by an amount equal to the mandatory Employee Contributions required by this Plan, and such Employee Contributions shall be contributed to the Plan by the Participating Employer on behalf of the Eligible Participant. Such Employee Contributions (referred to as Pick-Up Contributions) shall be treated as employer contributions for tax purposes pursuant to Code Section 414(h)(2).
- (b) Employee Contributions made under subsection (a) of this Section shall be paid from the same source of funds as used in paying wages to affected Employees. Employees do not have the option to receive the amounts contributed under subsection (a) directly.

- (c) The amounts of the contributions described in this Section shall be determined as follows:
  - (1) Five percent of Compensation each pay period until the Participant attains age thirty-five (35).
  - (2) Seven and half percent of Compensation each pay period from age thirty-five (35) through and including age forty-nine (49).
  - (3) Ten percent of Compensation each pay period after attaining age fifty (50).

#### 4.2 EMPLOYER CONTRIBUTIONS

This Plan shall provide for discretionary Employer Contributions. Unless and until this Plan is amended to the contrary, the discretionary Employer Contributions to a Participant's Account shall be in an amount as stated in this Section 4.2. For each pay period in which a Participant receives Compensation from a Participating Employer which results in Employee Contributions pursuant to Section 4.1, the Participating Employer shall contribute Employer Contributions in an amount that is precisely equal to the dollar amount of Employee Contributions contributed pursuant to Section 4.1 for the same pay period.

#### 4.3 ALLOCATION OF CONTRIBUTIONS AND EARNINGS

- (a) The Benefit Administrator shall establish and maintain an account in the name of each Participant (referred to as the Participant's Account) to which the Benefit Administrator shall credit as of each pay period all amounts allocated to each such Participant as hereafter set forth.
- (b) Employee Contributions and Employer Contributions made on or after January 1, 1998 shall be allocated to a Participant's Account as follows:
  - (1) With respect to the Participant's Employee Contributions made pursuant to Section 4.1(a), to the Participant's Employee Contributions Account (formerly called "Pick Up Account") in an amount equal to such Participant's mandatory Employee Contributions contributed since the last prior Valuation Date.
  - (2) With respect to the Employer Contributions made pursuant to Section 4.2 for a Participant, to the Participant's Employer Contributions Account (formerly called the "Non-Elective Account") in an amount equal to the Employer Contributions contributed since the last prior Valuation Date.
- (c) Each Participant shall have the right to direct the investment of all or a portion of his or her Participant Account balance. Each Participant may, direct the Benefit Administrator, to invest his or her Participant Account in specific Contracts or other investment funds permitted as investments under the Plan. Subject to direction by the Plan Administrator, the Benefit Administrator may establish a uniform, nondiscriminatory procedure and form for Participants use in providing this direction.

- (d) Expenses incurred in connection with a Participant's directed investment of the Participant's Account shall be charged against the balance in such Account.
- (e) A Participant may assign responsibility for investment elections and other transactions under the Plan to another party, in such manner as may be determined from time to time by the Plan Administrator and the Benefit Administrator.
- (f) Coincident with commencement of Employee and Employer Contributions (or a rollover or transfer into the Plan), a Participant shall make an investment direction election allocating his or her future contributions among one or more of the permissible investment funds or Contracts in increments of one percent (1%) or whole dollars. Such an investment election shall be presented in the manner, and in accordance with deadlines, established by the Plan Administrator. The Plan Administrator may establish and communicate that, in the absence of any such investment election by the Participant, all such contributions shall be invested in a default fund. In the absence of a timely investment election from the Participant, the Benefit Administrator shall instruct the Trustee to invest the Participant's contributions in such default fund until the effective date of any change in investment election by the Participant.
- (g) A Participant may change his or her investment elections with respect to future contributions, among one or more investment funds or Contracts, by giving direction in the manner, and in accordance with deadlines and limitations, established by the Plan Administrator and managed by the Benefit Administrator. The Participant's election shall specify a percentage in increments of one percent (1%), which percentage may not exceed one hundred percent (100%).
- (h) A Participant may re-invest existing balances of his or her Account, from one or more of the permissible investments, by giving direction in the manner, and in accordance with deadlines and limitations, established by the Plan Administrator and managed by the Benefit Administrator. Elections will be implemented in accordance with the written administrative procedures of the Plan and of the underlying investment fund, if any, including by way of example any limitations on the frequency of trading or transferring between investment funds. The Participant's transfer election shall specify either (i) a percentage in increments of one percent (1%), which aggregate percentage may not exceed one hundred percent (100%), or (ii) a dollar amount in whole dollars that is to be transferred.

#### 4.4 MAXIMUM ANNUAL ADDITIONS UNDER CODE SECTION 415

(a) Notwithstanding the foregoing, the maximum "annual additions" (as defined by Code Section 415(c)) credited to a Participant's Account for any "limitation year" shall equal the lesser of: (1) the applicable dollar limitation in effect for such "limitation year," as determined by the Secretary of Treasury, and as adjusted by the Secretary from time to time pursuant to Code Section 415(d)(1) (for example, \$53,000 for Plan Year 2015), or (2) 100% of the Participant's '415 Compensation' for such "limitation year." For any short "limitation year," the dollar limitation in (1) above shall be reduced by a fraction, the numerator of which is the number of full months in the short "limitation year" and the denominator of which is twelve (12).

- (b) For purposes of applying the limitations of Code Section 415, "annual additions" means the sum of any and all contributions or forfeitures credited to a Participant's Account under this Plan and any other defined contribution plan, for any "limitation year." "Annual additions" do not include transfers or rollovers of funds from one plan to another. In addition, the following are not "annual additions" for the purposes of this Section: (1) rollover contributions as defined in Code Sections 401(a)(31), 402(c)(1), 403(a)(4), 403(b)(8), 408(d)(3), and 457(e)(16); and (2) Employee contributions to a simplified employee pension excludable from gross income under Code Section 408(k)(6)."
- (c) For purposes of applying the limitations of Code Section 415, the "limitation year" shall be the Plan Year).
- (d) A Participant may not participate in more than one plan maintained by the Participating Employer pursuant to Code Section 401(a). Notwithstanding the foregoing, if a Participant does participate in more than one plan maintained by a Participating Employer pursuant to Code Section 401(a) which plans have different Anniversary Dates, the maximum "annual additions" under this Plan shall equal the maximum "annual additions" for the "limitation year" minus any "annual additions" previously credited to such Participant's accounts in any other plan during the "limitation year."
- (e) Notwithstanding anything contained in this Plan to the contrary, the limitations, adjustments and other requirements prescribed in this Plan shall at all times comply with the applicable provisions of Code Section 415 and the Regulations thereunder, as applicable to plans maintained by governmental entities, the terms of which are specifically incorporated herein by reference.

#### 4.5 TRANSFERS AND ROLLOVERS FROM ELIGIBLE RETIREMENT PLANS

- (a) Amounts may be transferred to the Plan Fund, by means of a direct trust-to-trust transfer or by means of a tax-free rollover from any other Eligible Retirement Plan (as that term is defined in Subsection 4.5(e) below); provided that the plan from which such funds are transferred permits the transfer to be made and provided that the Participant delivers information in a form and manner required by the Benefit Administrator to determine that the transfer or rollover is permissible under applicable law, subject to direction by the Plan Administrator. The amounts transferred shall be set up in a separate account herein referred to as a "Participant's Rollover Account." Such account shall be fully Vested at all times and shall not be subject to forfeiture for any reason.
- (b) Amounts in a Participant's Rollover Account shall be held in the Plan Fund pursuant to the provisions of this Plan, and may not be withdrawn by, or distributed to the Participant, in whole or in part, except as provided in this Plan.

- (c) At Normal Retirement Date, or such other date when the Participant or his Beneficiary shall be entitled to receive benefits, the fair market value of the Participant's Rollover Account shall be used to provide additional benefits to the Participant.
- (d) For purposes of this Section the term "amounts transferred from another plan" shall mean:
  - (1) amounts transferred to this Plan directly from another qualified plan;
  - (2) distributions from another qualified plan which are eligible rollover distributions and which are either transferred by the Employee to this Plan within sixty (60) days following their receipt thereof or are transferred pursuant to a direct rollover;
  - (3) amounts transferred to this Plan from a conduit individual retirement account. Prior to accepting any transfers to which this Section applies, the Benefit Administrator, subject to direction by the Plan Administrator, may require the Employee to establish and provide information supporting that the amounts to be transferred to this Plan meet the requirements of this Section.
- (e) For purposes of this Section, the term "Eligible Retirement Plan" shall mean any retirement plan that is qualified under Code Section 401(a) or 403(a), or, on or after January 1, 2002, which is designed to comply with the requirements of Code Section 403(b) or 457. Furthermore, "Eligible Retirement Plan" shall also mean a conduit individual retirement annuity or individual retirement account which is designed to comply with the requirements of Code Section 408(a) or 408(b), and which has no assets other than assets which: (i) were previously distributed to the Employee by another eligible retirement plan as a lump-sum distribution, (ii) were eligible for tax-free rollover to an eligible retirement plan, or (iii) were deposited in such conduit individual retirement account within sixty (60) days of receipt thereof and other than earnings on said assets, and (iv) amounts distributed to the Employee from a conduit individual retirement account meeting the requirements of clause (d)(3) above, and transferred by the Employee to this Plan within sixty (60) days of their receipt thereof from such conduit individual retirement account.

#### 4.6 ADJUSTMENT FOR EXCESSIVE ANNUAL ADDITIONS

- (a) For "limitation years" beginning on or after July 1, 2007, if the Benefit Administrator determines that the amount of a Participant's "annual additions" exceed the limitations set forth herein, the excess "annual additions" shall be corrected as permitted under the Internal Revenue Service's Employee Plans Compliance Resolution System ("EPCRS"), including any successor system or program thereto, or pursuant to any other available guidance from the Internal Revenue Service or U.S. Department of the Treasury.
- (b) If, as a result of a reasonable error in estimating a Participant's Compensation, an error is made in determining the amount of Participant Pick Up Contributions that may be made with respect to any individual under the limits of Code Section 415, the "annual additions" under this

Plan would cause the maximum 'annual additions' to be exceeded for any Participant, the Benefit Administrator, to the extent permitted by paragraph (a) of this Section 4.6, shall return Employee Contributions attributable to the affected Participant. To the extent permitted by paragraph (a) of this Section 4.6, any Employer Contributions attributable to such return shall be drawn from the Participant's Account and shall be held in suspense, to be used to reduce future Employer Contributions to the Plan.

#### 5. ACCOUNTS, INVESTMENTS AND VALUATIONS

#### 5.1 INDIVIDUAL ACCOUNTS

The Benefit Administrator shall establish and maintain an Account for each Participant that shall reflect contributions made on behalf of the Participant, and earnings, expenses, gains and losses attributable thereto, and investments made with amounts in the Participant's Account.

#### 5.2 ALLOCATION OF EARNINGS AND LOSSES TO ACCOUNTS

As of each Valuation Date, the Benefit Administrator shall:

- (a) Adjust the balances in each of the Participant's Accounts upward or downward to reflect investment gains and losses (adjusted by any expenses charged to the Plan that are charged to the Participant's Account) since the last Valuation Date; the gain or loss of each separate investment fund or Contract will be allocated to each Participant's Account in the same proportion that the value of Participant's investment in such fund or Contract as of the last Valuation Date bears to the value of all Participant Accounts invested in that fund or Contract as of the same date;
- (b) Credit to the Accounts of each Participant the contributions which were deposited in the Plan Fund since the last Valuation Date;
- (c) Charge to each Participant's Account all payments or distributions made to or on behalf of the Participant since the last Valuation Date;
- (d) Charge each Participant's Account such Participant's share of any administrative expenses as are allocable to such Account; and
- (e) Credit and charge the proper investment fund or Contract subaccount of each Participant to reflect transfers among investment funds.

#### 5.3 ACCOUNT STATEMENTS

Each Participant shall be provided by the Benefit Administrator with a statement of his or her Account under the Plan showing the values of each investment fund or Contract, and the

aggregate value, at least once each Plan Year. Account statements may be delivered in hard copy or electronically as determined by the Plan Administrator.

# 5.4 PERMISSIBLE INVESTMENTS

On and after the Restatement Effective Date, the State Board shall arrange for the establishment of one or more investment funds and Contracts within the Trust Fund. The State Board may change the investment funds and Contracts from time to time in its discretion. The State Board shall have the authority and responsibility to select, eliminate or replace the investment funds and Contracts for the Plan, and to monitor the investment performance of the funds and Contracts offered under the Plan. In exercising this authority, the State Board will consider the advice and recommendations of the Investment Advisory Committee.

Except as otherwise provided in this Article 5, all contributions allocated to a Participant's Account shall be invested in one or more of the permissible investments, in accordance with the investment elections of the Participant (or, where applicable, the alternate payee or beneficiary with rights over an Account). As is the case for an active Participant, an inactive Participant may continue to make investment elections with respect to his or her Account.

#### 5.5 PURPOSE OF THE PLAN FUND

A Plan Fund has been created and will be maintained for the purposes of the Plan, consisting of:

- (a) one or more Contracts which shall have the status of qualified trust assets of the Plan in accordance with Code Section 401(f), and
- (b) on and after the Restatement Date, trust assets held by the Trustee in accordance with the terms of a trust agreement between the Sponsor and the Trustee.

The Trustee shall be appointed by the State Board, as further provided in Article 2 of this Plan. All contributions will be paid into the Plan Fund, and all benefits under the Plan will be paid from the Trust.

#### 5.6 EXCLUSIVE BENEFIT OF PARTICIPANTS AND BENEFICIARIES

Subject to the provisions for the return of erroneous contributions, the Plan Fund will be used and applied only in accordance with the provisions of the Plan to provide the benefits thereof, and no part of the corpus or income of the Plan Fund shall be used for or diverted to purposes other than for the exclusive benefit of the Participants and their Beneficiaries and with respect to expenses of administration.

#### 5.7 BENEFITS SUPPORTED ONLY BY THE PLAN FUND

Any person having any claim under the Plan will look solely to the assets of the Plan Fund for satisfaction; provided, however, that any Supplemental Retirement Benefit payable pursuant to

provisions of the State Board Supplemental Retirement Plan Document shall not be paid from the Plan Fund established for this Plan.

#### 5.8 PAYMENT OF EXPENSES

Except as provided below, all reasonable costs and expenses incident to the administration and protection of the Plan and the Plan Fund, including but not limited to legal, accounting, and Trustee fees, shall be paid by the Trust, unless the Plan Administrator elects to cause the State Board or any or all of the Participating Employers to pay such expenses. Until paid, such expenses shall constitute a first and prior claim and lien against the Plan Fund.

All such costs and expenses paid from the Plan Fund, unless allocable to one or more Participant Accounts for transaction expenses attributable directly to such Participants, shall be charged against the Accounts of all Participants on a pro rata basis or in such other reasonable manner as may be directed by the Plan Administrator.

Notwithstanding the foregoing, any and all expenses relating to settlor functions that arise from the creation, design or termination of the Plan must be paid by the Sponsor or Participating Employers and may not be paid from the Plan Fund.

#### 6. DETERMINATION AND DISTRIBUTION OF BENEFITS

#### 6.1 DETERMINATION OF BENEFITS UPON RETIREMENT

A Participant may elect to have Retirement Benefits commence at any time after he or she retires under the Plan; 90 days after his or her employment with any and all Participating Employer(s) terminates; or, after he or she has been approved to begin participation in the "phased retirement" program by his or her Participating Employer and pursuant to the criteria established in Section 6.1(a).

Upon such benefit commencement date, all amounts credited to such Participant's Account shall become eligible for distribution in accordance with the form of payment elected by the Participant. However, subject to the rules pertaining to mandatory commencement of distributions under Article 11 and Code Section 401(a)(9), a Participant may postpone the commencement of benefit payments under this Plan. Upon a Participant's Normal or Late Retirement Date (whichever occurs first), or as soon thereafter as is practicable, the Benefit Administrator shall distribute all amounts credited to such Participant's Account in accordance with Section 6.3.

- (a) Phased retirement enables Plan Participants with full time appointments to transition into retirement through working part-time. "Phased Retirement" as used in Section 6.1 applies to:
  - (1) Participants age 59 ½ or older with at least 10 years of full-time service in the Plan; and

- (2) Who terminate from their positions with a Participating Employer, surrendering tenure or rights to continuous appointment; and
- (3) Who are rehired, within six months of termination from their position, by a Participating Employer into a different job and assigned a reduced work load. A Retired Participant rehired after a two consecutive quarters or a six continuous month break in employment from all Participating Employers is considered to have had a complete separation of service from a Participating Employer.

Participation in phased retirement is completely voluntary and is not an absolute right of a Plan Participant. Phased retirement is only available when agreed to and entered into by a mutual agreement between an eligible Participant and his or her Participating Employer. From a Plan perspective, Participants in the phased retirement program are not considered to have retired or separated from employment, but remain eligible for the Participating Employer's contribution to the Plan. The Participating Employer will continue to make contributions on behalf of the Participant to the Plan as delineated in Article 4.

Participation in the Plan's phased retirement program is not intended to impact eligibility for other employment benefits that may be offered to Plan Participants through their Participating Employer or by the State of Washington.

#### 6.2 DETERMINATION OF BENEFITS UPON TERMINATION

- (a) On or before the Anniversary Date coinciding with or subsequent to the termination of a Participant's employment for any reason other than death or retirement, the Plan Administrator may direct the Benefit Administrator to segregate a Terminated Participant's Account and invest the aggregate amount thereof in a separate annuity Contract. In the event the Participant's Account is not segregated, the amount shall remain in a separate account for the Terminated Participant and share in investment gains and losses per Section 5.2 until such time as a distribution is made to the Terminated Participant.
- (b) Distribution of the funds due to a Terminated Participant shall be made to Participants who have either: (i) attained age fifty-five (55), (ii) separated from service due to disability, (iii) completed thirty Years of Service, or (iv) been separated from service from all Participating Employers for more than 90 days. Distribution of funds may also occur to Participants who are participating in the Phased Retirement program as described in Section 6.1. Any distribution under this paragraph shall be made in a manner that is consistent with and satisfies the provisions of Section 6.3, including, but not limited to, all notice and consent requirements of Code Sections 411(a)(11) and 417.

For purposes of this Section 6.2, if the value of a Terminated Participant's benefit is zero, the Terminated Participant shall be deemed to have received a distribution of such benefit.

- (c) Participants shall be fully Vested in their Employer Contributions and Employee Contributions immediately upon entry into the Plan. Consequently, the full balance of a Participant's Account shall be fully Vested at all times, and (except as provided under Section 3.6, with regard to erroneous contributions) shall not be subject to forfeiture under any circumstances.
- (d) The computation of a Participant's Vested percentage of his or her interest in the Plan shall not be reduced as the result of any direct or indirect amendment to this Plan. In the event that the Plan is amended to change or modify any vesting schedule, a Participant with at least three (3) Years of Service as of the expiration date of the election period may elect to have his or her Vested percentage computed under the Plan without regard to such amendment. If a Participant fails to make such election, then such Participant shall be subject to the new vesting schedule. The Participant's election period shall commence on the adoption date of the amendment and shall end 60 days after the latest of:
  - (1) the adoption date of the amendment,
  - (2) the effective date of the amendment, or
  - (3) the date the Participant receives written notice of the amendment and the end of the 60 day period from the Employer or Plan Administrator.
- (e) If any Former Participant shall be reemployed by the Participating Employer, the Participant shall continue to participate in the Plan in the same manner as if such termination had not occurred provided that he or she is re-employed in an eligible position and establishes eligibility pursuant to Section 3.1.

#### 6.3 DISTRIBUTION OF BENEFITS; SPOUSAL CONSENT

- (a) To begin receiving benefits, the Participant or Beneficiary must write directly to the Benefit Administrator, using the forms and process established by the Benefit Administrator. Distribution of benefits to a Participant as described in this Section requires consent of the Participant's spouse. The consent must be in writing, must acknowledge the effect of the election or action to which the consent applies and must be witnessed by a notary public. Unless the consent expressly provides that the Participant may make further elections without further consent of the spouse, the consent will be effective only with respect to the specific election of form of benefit or Beneficiary or both to which the consent relates. Spousal consent will be effective only with respect to that spouse. Spousal consent will not be required if it is established to the satisfaction of the Plan Administrator that the spouse cannot be located.
- (b) The Benefit Administrator, pursuant to the election of the Participant (or Designated Beneficiary), shall direct the distribution to a Participant (or the Participant's Designated Beneficiary) any amount to which he or she is entitled under the Plan in one or more of the following methods:

- (1) With respect to any portion of the balance of a Participant's Account that is held in an investment fund other than a Contract, the optional forms of distribution shall be as follows:
  - (i) one lump-sum payment in cash;
  - (ii) systematic distributions, consisting of distributions in an amount per payment elected by the Participant, and occurring either semi-monthly, monthly, quarterly, semi-annually or annually (as elected by the Participant), and continuing until the earliest to occur of the date: (A) the Participant directs the distributions to cease; (B) the Participant directs the amount of each distribution to change; (C) the exhaustion of the balance of the Participant's Account; or (D) the death of the Participant (and, in case of such a death, any remaining balance shall be payable to the Participant's Designated Beneficiary); or
  - (iii) upon the reinvestment transfer to a Contract of any or all of the balance of the Participant's Account that is held in an investment fund, then any such transferred balance that is held in an investment vehicle under a Contract shall thereafter be eligible for distribution in accordance with any of the alternate forms of distribution stated in subsection "(2)" below.
- (2) With respect to any portion of the balance of a Participant's Account that is held in an investment vehicle under a Contract, the optional forms of distribution shall be as follows:
  - (i) One lump sum payment in cash. Under the Group Retirement Annuity (GRA) contract, cash payments from the TIAA Traditional Annuity must be made no later than 120 days after the 90 day period referenced in Section 6.1 expires. Alternatively, lump-sum withdrawals from the TIAA Traditional Annuity may be received through the Transfer Payout Annuity in a stream of 10 payments over nine years and one day. There is no time limit on cash payments from the CREF and TIAA Real Estate Accounts.
  - (ii) Purchase of or providing an annuity in a form selected by the Participant, acceptable to the Benefit Administrator and provided under the Contract, or
  - (iii) Payments over a period certain in monthly, quarterly, semiannual, or annual cash installments according to terms selected by the Participant acceptable to the Benefit Administrator and provided under the Contract.
  - (iv) In the form of a Retirement Transition Benefit. The Retirement Transition Benefit ("RTB") permits a Participant to receive a one-time lump-sum payment for up to 10 percent of his or her Participant Account at the time annuity income begins, provided the one-sum payment from each Contract does not exceed 10 percent of the Participant Account being converted to retirement income.

- (v) In the form of an Interest Payment Retirement Option. Participants who are age 55 or older may receive payments of the interest that would otherwise be credited to their TIAA Traditional Annuity accumulation.
- (vi) A systematic distribution, as described in Section 6.3(b)(1)(ii).
- (b) Any distribution to a Participant shall require such Participant's consent. With regard to this required consent:
  - (1) No consent shall be valid unless the Participant has received a general description of the material features and an explanation of the relative values of the optional forms of benefit available under the Plan that would satisfy the notice requirements of Code Section 417.
  - (2) A Terminated Participant who has a right to commence a benefit prior to his or her Normal Retirement Date must be informed of his or her right to defer commencement of the benefit to a subsequent date, but not later than the date required by Article 11. If a Terminated Participant fails to consent to a distribution, it shall be deemed an election to defer the commencement of payment of any benefit. However, any election to defer the receipt of benefits shall not apply with respect to distributions that are required to commence under Article 11.
  - (3) Notice of the rights specified under this paragraph shall be provided no less than 30 days and no more than 90 days before the "annuity starting date" (as defined in the Code).
  - (4) Written consent of the Participant to the distribution must not be made before the Participant receives the notice and must not be made more than 90 days before the first day on which all events have occurred which entitle the Participant to such benefit.
  - (5) No consent shall be valid if a significant detriment is imposed under the Plan on any Participant who does not consent to the distribution.
  - (6) If a distribution is one to which Code Sections 401(a)(11) and 417 do not apply, such distribution may commence less than 30 days after the notice required under Regulation 1.411(a)-11(c) is given, provided that: (1) the Benefit Administrator clearly informs the Participant that the Participant has a right to a period of at least 30 days after receiving the notice to consider the decision of whether or not to elect a distribution (and, if applicable, a particular distribution option), and (2) the Participant, after receiving the notice, affirmatively elects a distribution.

#### 6.4 DISTRIBUTION OF BENEFITS UPON DEATH

(a) If a Participant dies before he or she has begun to receive distribution of his or her benefits under the Plan, the Benefit Administrator shall direct that the Participant's Designated Beneficiary be paid all amounts credited to such Participant's Account which are payable to the Designated Beneficiary under the Plan, in one or more of the methods described in Section 6.3.

- (b) Any distribution to a Participant or Designated Beneficiary must meet the requirements of Code Section 401(a)(9) as set forth in Article 11.
  - (1) If the distribution of a Participant's interest has begun (or is deemed to have begun in accordance with the Code) and the Participant dies before his or her entire interest has been distributed, the remaining portion of such interest shall be distributed at least as rapidly as under the method of distribution selected pursuant to Section 6.3 as of the Participant's date of death.
  - (2) If a Participant dies before he or she has begun to receive any distributions of his or her interest under the Plan (or before distributions are deemed to have begun pursuant to the Code), then the Participant's death benefit shall be distributed to his or her Beneficiaries in accordance with the following:
    - (i) The entire death benefit shall be distributed to the Participant's Beneficiaries by December 31st of the calendar year immediately following the calendar year in which the fifth anniversary of the Participant's death occurs.
    - (ii) The 5-year distribution requirement of (i) above shall not apply to any portion of the deceased Participant's interest that is payable to or for the benefit of a Designated Beneficiary. In such event, such portion shall be distributed over the life of such Designated Beneficiary (or over a period not extending beyond the life expectancy of such Designated Beneficiary) provided such distribution begins not later than December 31st of the calendar year immediately following the calendar year in which the Participant died.
    - (iii) However, in the event the Participant's spouse (determined as of the date of the Participant's death) is the Designated Beneficiary, a spousal election may be made to delay the time at which distributions must commence until the later of: (A) December 31st of the calendar year immediately following the calendar year in which the Participant died; or (B) December 31st of the calendar year in which the Participant would have attained age 70 1/2. If the surviving spouse dies before distributions to such spouse begin, then the 5-year distribution requirement of this Section shall apply as if the spouse was the Participant.
- (c) For the purpose of Section 6.4(b), the election by a Designated Beneficiary to be excepted from the 5-year distribution requirement must be made no later than December 31st of the calendar year following the calendar year of the Participant's death. Except, however, with respect to a Designated Beneficiary who is the Participant's surviving spouse, the election by the Participant's surviving spouse to delay the payment of benefits must be made by the earlier of: (1) December 31st of the calendar year immediately following the calendar year in which the Participant died or, if later, the calendar year in which the Participant would have attained age 70 1/2; or (2) December 31st of the calendar year that contains the fifth anniversary of the date of the Participant's death. An election by a Designated Beneficiary must be in writing and shall be irrevocable as of the last day of

the election period state herein. In the absence of an election by the Participant or a Designated Beneficiary, the 5-year distribution requirement shall apply.

For purposes of this Section, the life expectancy of a Participant and a Participant's spouse (other than in the case of a life annuity) shall be re-determined annually in accordance with Regulations. Life expectancy and joint and last survivor expectancy shall be computed using the return multiples in Tables V and VI of Regulation Section 1.72-9.

#### 6.5 DISTRIBUTION FOR MINOR BENEFICIARY

In the event a distribution is to be made to a minor, then the Benefit Administrator may direct that such distribution be paid to the legal guardian, or if none, to a parent of such Designated Beneficiary or a responsible adult with whom the Designated Beneficiary maintains his residence, or to the custodian for such Designated Beneficiary under the Uniform Gift to Minors Act or Gift to Minors Act, if such is permitted by the laws of the state in which said Designated Beneficiary resides. Such a payment to the legal guardian, custodian or parent of a minor Designated Beneficiary shall fully discharge the Sponsor, Plan Administrator, Benefit Administrator, Participating Employer, and Plan from further liability on account thereof.

# 6.6 LOCATION OF PARTICIPANT OR BENEFICIARY UNKNOWN

In the event that all, or any portion, of the distribution payable to a Participant or Designated Beneficiary hereunder shall, at the Participant's attainment of age 70-1/2, remain unpaid solely by reason of the inability of the Benefit Administrator, after sending a registered letter, return receipt requested, to the last known address, and after further diligent effort, to ascertain the whereabouts of such Participant or Designated Beneficiary, the amount so distributable shall be disposed of pursuant to state law. This effort and distribution are subject to direction of the Plan Administrator.

#### 6.7 DETERMINATION OF BENEFITS IN EVENT OF DISABILITY

- (a) A Participating Employer may approve the retirement of any Participant who has terminated employment for reasons of health or permanent disability, either upon the request of the Appointing Authority or the Participant, provided that reasonable consideration is first given to the written recommendations of the Employee's personal physician or, if requested by the Participant or the Appointing Authority, a review of such recommendations by another physician appointed by mutual agreement for that purpose.
- (b) The Participant described in subsection (a) shall be treated in all respects as if he or she had attained his or her Normal Retirement Date. As of the date that would be the Participant's Normal Retirement Date, the Benefit Administrator, in accordance with the provisions of Section 6.3, shall direct the Custodian to distribute to such Participant all amounts credited to such Participant's Account as though the Participant had retired.

#### 6.8 ADVANCE DISTRIBUTION FOR HARDSHIP

- (a) In the event of a Participant's financial hardship, as such term is described herein, a Participant shall have a right to withdraw from his or her Account an amount up to the lesser of the Account balance or the amount necessary to satisfy the immediate and heavy financial need of the Participant; provided, however, such a withdrawal shall be limited to the Participant's Deferred Compensation and any income allocable thereto credited to the Participant's Account prior to December 31, 1988. Withdrawal under this Section shall be authorized only if the distribution is as a result of:
  - (1) Expenses for (or necessary to obtain) medical care that would be deductible under Code Section 213(d) (determined without regard to whether the expenses exceed 7.5% of adjusted gross income) incurred by the Participant, his or her spouse, children, or dependents;
  - (2) The need to prevent the eviction of the Participant from the Participant's principal residence or foreclosure on the mortgage of the Participant's principal residence; or
  - (3) The need to prevent the bankruptcy of the Participant
- (b) No distribution shall be made pursuant to this Section unless the Participating Employer and Benefit Administrator, based upon the Participant's representation and such other facts as are known to the Participating Employer, determine that all of the following conditions are satisfied:
  - (1) The distribution is not in excess of the amount of the immediate and heavy financial need of the Participant, where the amount of the immediate and heavy financial need may include any amounts necessary to pay any federal, state or local income taxes or penalties reasonably anticipated to result from such distribution;
  - (2) The Participant has obtained all distributions, other than hardship distributions, and all nontaxable loans currently available under all plans maintained by the Participating Employer; and
  - (3) The Plan, and all other plans maintained by the Participating Employer, provide that the Participant's elective deferrals to any other retirement plans (including, by way of example, programs under Code Section 403(b) or 457) will be suspended for at least six months after receipt of the hardship distribution.
- (c) Any distribution made pursuant to this paragraph shall be deemed to be made as of the first day of the Plan Year or, if later, the Valuation Date immediately preceding the date of distribution, and the Participant's Account shall be adjusted accordingly.
- (d) Any distribution made pursuant to this Section 6.8 shall be subject to the consent of the Participant's spouse, if any, and shall be paid as a single distribution.

- (e) The Participant shall be deemed to have no other resources reasonably available to meet the need if the Participant certifies that he or she cannot meet the need through:
  - (1) Reimbursement or compensation by insurance or another source;
  - (2) Reasonable liquidation of assets;
  - (3) Borrowing from supplemental retirement accounts, life insurance values, or commercial sources; and/or
  - (4) Stopping any voluntary employee contributions to tax deferral or savings plans made available to by the employer, provided however that mandatory employee contributions to the Plan must continue while the Participant remains eligible for the Plan.

# 6.9 QUALIFIED DOMESTIC RELATIONS ORDERS

All rights and benefits, including elections, provided to a Participant in this Plan shall be subject to the rights afforded to any "alternate payee" under a "qualified domestic relations order." Furthermore, a distribution to an "alternate payee" shall be permitted if such distribution is authorized by a "qualified domestic relations order," even if the affected Participant has not reached the "earliest retirement age" under the Plan. For the purposes of this Section, "alternate payee," "qualified domestic relations order" and "earliest retirement age" shall have the meaning set forth under Code Section 414(p).

#### 6.10 DIRECT ROLLOVER

- (a) Notwithstanding any provision of the Plan to the contrary that would otherwise limit a distributee's election under this Section, a distributee may elect, at the time and in the manner prescribed by the Benefit Administrator, to have any portion of an eligible rollover distribution paid directly to an Eligible Retirement Plan specified by the distributee in a direct rollover.
- (b) For purposes of this Section the following definitions shall apply:
  - (1) Eligible rollover distribution: An eligible rollover distribution is any distribution of all or any portion of the balance to the credit of the distributee, except that an eligible rollover distribution does not include: any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for life (or life expectancy) of the distributee or the joint lives (or joint life expectancies) of the distributee and the distributee's Designated Beneficiary, or for a specified period of ten years or more; any minimum required distribution under Code Section 401(a)(9); any distribution made on account of hardship; and (except as permitted by the Code) any distribution that is not includible in gross income.
  - (2) Eligible Retirement Plan shall mean an individual retirement account or individual retirement annuity ("IRA"), other than an endowment contract, under Code

Section 408(a) or 408(b), a Roth IRA under Code Section 408A, a trust qualified under Code Section 401(a) and exempt from tax under Code Section 501(a), which accepts rollover distributions (as limited by Code Section 401(a)(31)(D)), a plan designed to comply with Code Section 403(a) or 403(b), or an eligible plan under Code Section 457 that is maintained by a governmental entity which agrees to separately account for amounts transferred into such plan from this Plan. In the case of a distribution to a Beneficiary who at the time of the Participant's death was neither the spouse of the Participant nor the spouse or former spouse of the Participant who is an alternate payee under a domestic relations order, a direct rollover is payable only to an IRA that has been established on behalf of the Beneficiary as an inherited IRA (within the meaning of Code Section 408(d)(3)(C)).

- (3) A distributee includes a Participant or Former Participant, the Participant's or Former Participant's surviving spouse, the Participant's or Former Participant's spouse or former spouse who is the alternate payee under a qualified domestic relations order, as defined in Code Section 414(p), and the non-spouse Beneficiary of the Participant or Former Participant.
- (4) A direct rollover is a payment by the Plan to the Eligible Retirement Plan specified by the distributee.

#### 7. PLAN ADMINISTRATRATION

#### 7.1 PLAN ADMINISTRATOR

- (a) Effective on and after the Restatement Effective Date, unless and until the State Board determines to the contrary, the Plan Administrator shall be the Deputy Executive Director of Business Operations of the State Board.
- (b) The Plan Administrator shall have the authority to appoint, replace and terminate the service of a legal counsel, consultant or adviser who may be selected to advise the Plan Administrator.
- (c) The Plan Administrator shall have all powers necessary to enable it to carry out its duties under the Plan. Not in limitation, but in amplification of the foregoing, it will have the discretionary power to construe the Plan and to determine all questions that may arise hereunder, including all questions relating to the eligibility of Employees to participate in the Plan and the amount of benefit to which any Participant or Designated Beneficiary may become entitled. Decisions upon all matters within the scope of his or her authority will be final.
- (d) The Plan Administrator, with the assistance and contractual services of the Benefit Administrator, shall administer the Plan in a nondiscriminatory manner for the exclusive benefit of Participants and their Beneficiaries. The Plan Administrator (or the persons or entities to whom the Plan Administrator delegates any of its discretion, authority, duties and responsibilities, including the Benefit Administrator) shall have the discretion in the performance

of – and shall perform – all such duties as are necessary to supervise the administration of the Plan and to control its operation in accordance with the terms thereof, including, but not limited to, the following:

- (1) Make and enforce such rules and regulations as it shall deem necessary or proper for the efficient administration of the Plan, consistent with State Board policies;
- (2) Interpret the provisions of the Plan and resolve any question arising under the Plan, or in connection with the administration or operation thereof;
- (3) Make all determinations affecting the eligibility of any Employee to be or become a Participant;
- (4) Determine eligibility for and amount of benefits for any Participant;
- (5) Authorize and direct the Trustee with respect to all disbursements of benefits under the Plan;
- (6) Employ and engage such persons, counsel and agents and to obtain such administrative, consulting, clerical, medical, legal, audit and actuarial services as it may deem necessary in carrying out the provisions of the Plan, including without limitation, the Benefit Administrator; and
- (7) Delegate and allocate specific responsibilities, obligations and duties imposed by the Plan to one or more employees, officers or such other persons as the Plan Administrator deems appropriate, including the Benefit Administrator and the Trustee.

#### 7.2 INVESTMENT ADVISORY COMMITTEE; STATE BOARD POWERS

- (a) The Investment Advisory Committee may consist of up to seven employees of the Sponsor or of a Participating Employer, including, without limitation, the Deputy Executive Director of Business Operations of the Sponsor, who shall serve as the voting chairperson of the Investment Advisory Committee. The chairperson shall have the authority to appoint, remove and replace other voting members of the Investment Advisory Committee. Any member of the Investment Advisory Committee may resign or may be removed by the chairperson of the Investment Advisory Committee.
- (b) The Investment Advisory Committee shall serve as an advisory committee to the State Board and Plan Administrator. The State Board (with the consideration of advice of the Investment Advisory Committee) shall have the following powers and duties with respect to Contracts as delineated in the Investment Policy Statement, other investment fund options offered by the Plan, and any Investment Managers:
  - (1) To set policy for the investment vehicles to be offered to Participant's for investment direction under the Plan Fund:

- (2) To review, monitor and advise selection or removal of investment funds and Contracts:
- (3) To receive and review reports of the performance of investment funds and Contracts;
- (4) To review and advise selection or removal of one or more Investment Managers;
- (5) To interpret and construe the Plan with respect to the investment, reinvestment, and disposition of Plan assets; and
- (6) To appoint, replace and terminate an investment consultant or advisor who may be selected to advise the Investment Advisory Committee.

## 7.3 INDEMNITY, BONDING AND INSURANCE

- (a) The Sponsor shall indemnify and hold harmless the Plan Administrator, from and against any and all loss resulting from liability to which the Plan Administrator may be subjected by reason of any act or conduct (except willful or reckless misconduct), in its official capacities in the administration of the Plan, or the Plan Fund, or both, including all expenses reasonably incurred in the Plan Administrator's defense, in case the Sponsor fails to provide such defense.
- (b) To the extent required by law, every member, every fiduciary of the Plan and every person handling Plan funds shall be bonded. The Plan Administrator shall take such steps as are necessary to assure compliance with applicable bonding requirements. The Plan Administrator may apply for and obtain fiduciary liability insurance insuring the Plan against damages by reason of breach of fiduciary responsibility at the Plan's expense and insuring each fiduciary against liability to the extent permissible by law. Nothing in this paragraph is intended to impose any duty upon the State Board or the Plan Administrator to assume the cost of insurance coverage, or to indemnify, any Benefit Administrator, Trustee or any other third-party advisor, fiduciary or service provider.

### 7.4 FIDUCIARIES

- (a) Limitation of Liability: To the extent permitted by law, no Participant shall have any claim against the Sponsor, any Participating Employer, the Plan Administrator, the Investment Advisory Committee members, the Trustee or any other fiduciary or service provider to the Plan, or against the directors, officers, members, agents or representatives of any of them (collectively "parties in interest"), for any benefits under the Plan, and such benefits (other than those described in Section 6.2) shall be payable solely from the Trust Fund; nor shall any of the parties in interest incur any liability to any person for any action taken or suffered or omitted to be taken by them under the Plan in good faith. The Sponsor intends that neither the Plan nor any party in interest shall be liable for any loss due to a Participant exercising control over the investment of assets in his or her Account.
- (b) Indemnification

- (1) Parties Protected by Indemnification. In order to facilitate the recruitment of competent individuals as fiduciaries and/or service providers to the Plan, the Sponsor agrees to provide the indemnification as described herein. This provision shall apply to Employees who are considered Plan fiduciaries or service providers to the Plan. Notwithstanding the preceding, this provision shall not apply and indemnification will not be provided for any third party corporation, partnership or other business entity that may be appointed as a fiduciary or service provider to the Plan, or to any corporate Benefit Administrator, Trustee or Investment Manager, whether appointed as a service provider or a Plan fiduciary.
- (2) Scope of Indemnification. The Sponsor agrees to indemnify an Employee fiduciary as described above for all acts taken in good faith in carrying out his or her responsibilities under the terms of this Plan or other responsibilities imposed upon such fiduciary by applicable state law. This indemnification for all acts is intentionally broad but shall not provide indemnification for gross negligence, willful misconduct, embezzlement or diversion of Plan assets for the benefit of the Employee fiduciary. The Sponsor agrees to indemnify Employees and other individuals as described in the previous paragraph herein for all expenses of defending an action by a Participant, Designated Beneficiary or government entity, including all legal fees for counsel selected with the consent of the Sponsor and the Participating Employers and other costs of such defense. The Sponsor and the Participating Employers will also reimburse an Employee fiduciary for any monetary recovery in any court or arbitration proceeding. In addition, if the claim is settled out of court with the concurrence of the Sponsor and the Participating Employer, the Sponsor and Participating Employer will indemnify an Employee fiduciary for any monetary liability under said settlement. The Sponsor and the Participating Employer shall have the right, but not the obligation, to conduct the defense of such persons in any proceeding to which this Section applies. The Sponsor may satisfy its obligations under this Section in whole or in part through the purchase of a policy or policies of insurance providing equivalent protection.

### 7.5 CLAIMS AND APPEALS PROCEDURES

- (a) The claims and appeals procedures under the Plan shall be administered in accordance with the provisions of this Section 7.5 and any guidelines adopted by the Plan Administrator from time to time that comply with applicable law.
- (b) A "claim" (as that term is used in this Section) occurs when a Participant or Designated Beneficiary ("Claimant") either (i) makes an application for a benefit under the Plan, or (ii) disputes a determination by a Participating Employer or the Benefit Administrator (or a person or persons authorized by the Plan Administrator to process claims hereunder) of the amount of any benefit or the resolution of any matter affecting a benefit under the Plan. A claim or appeal may be filed by an authorized representative of the Claimant.

- (c) Notwithstanding any other provision of the Plan, a Claimant shall not have a right to submit a dispute with respect to a benefit under this Plan more than one year after the date the individual has knowledge of all material facts that are the subject of the dispute.
- (d) Claims for benefits under the Plan shall be filed with the office of the Deputy Executive Director of Business Operations of the State Board on forms or in a manner provided for that purpose. Each claim will be decided by a person appointed by the Plan Administrator to serve as a claims administrator under this Plan (the "Claims Administrator"). The Claimant will be notified in writing of the approval or denial of a claim within ninety (90) days of the date that the Claims Administrator receives the claim, unless special circumstances require an extension of time for processing the claim. In the event an extension is necessary, the Claimant will be provided written notice prior to the end of the initial ninety (90) day period indicating the special circumstances requiring the extension and the date by which the Claims Administrator expects to notify the Claimant of approval or denial of the claim. In no event will an extension extend beyond ninety (90) days after the end of the initial ninety (90) day period.
- (e) If a claim is denied, the written notification will state specific reasons for the denial, make specific reference to the Plan provision(s) on which the denial is based, and provide a description of any material or information necessary for the Claimant to perfect the claim and why such material or information is necessary. The written notification will also provide a description of the Plan's review procedures and the applicable time limits.
- The Claimant will have sixty (60) days from receipt of the written notification of the denial of a claim to file a signed, written request for a full and fair review of the denial by the Plan Administrator, who shall serve as an appeals administrator for this purpose (the "Appeals Administrator"). This request should include the reasons the Claimant is requesting a review and may include facts supporting the request and any other relevant comments, documents, records and other information relating to the claim. Upon request and free of charge, the Claimant will be provided with reasonable access to, and copies of, all documents, records and other information relevant to the claim, including any document, record or other information that was relied upon in, or submitted, considered or generated in the course of, denying the claim. A final, written determination of the Claimant's eligibility for benefits shall be made within sixty (60) days of receipt of the Claimant's request for review, unless special circumstances require an extension of time for processing the claim, in which case the Claimant will be provided written notice of the reasons for the delay within the initial sixty (60) day period and the date by which the Claimant should expect notification of approval or denial of the claim. This review will take into account all comments, documents, records and other information submitted by the Claimant relating to the claim, whether or not submitted or considered in the initial review of the claim. In no event will an extension extend beyond sixty (60) days after the end of the initial sixty (60) day period. If an extension is required because the Claimant fails to submit information that is necessary to decide the claim, the period for making the benefit determination on review will be tolled from the date the notice of extension is sent to the Claimant until the date on which the Claimant responds to the request for additional information. If the claim is denied on review, the written notification will state specific reasons for the denial, make specific reference to the Plan provision(s) on which the denial is based and state that the Claimant is entitled to receive upon

request, and free of charge, reasonable access to, and copies of, all documents, records and other information relevant to the claim, including any document, record or other information that was relied upon in, or submitted, considered or generated in the course of, denying the claim.

- (g) If the claim is initially denied or is denied upon review, the Claimant shall be entitled to receive upon request, and free of charge, reasonable access to, and copies of, any document, record or other information that demonstrates that (1) the claim was denied in accordance with the terms of the Plan, and (2) the provisions of the Plan have been consistently applied to similarly situated Participants, if any. In pursuing any of its rights set forth in this Section, a Claimant's authorized representative may act on the Claimant's behalf.
- (h) The Plan hereby delegates full and complete discretion to the Claims Administrator and the Appeals Administrator:
  - (1) To make findings of fact pertaining to a claim or appeal;
  - (2) To interpret the Plan as applied to the facts; and
  - (3) To decide all aspects of the claim or appeal.

The decision by the Appeals Administrator shall be the final and conclusive administrative review proceeding under the Plan.

## 8. AMENDMENT, TERMINATION AND MERGERS

## 8.1 AMENDMENT

- (a) The Sponsor shall have the right at any time to amend the Plan and to delegate such authority to the Plan Administrator. Any such amendment or delegation shall be adopted by formal action of the State Board and executed by the Plan Administrator or other person authorized to act on behalf of the State Board. Any such amendment shall become effective as provided therein upon its execution.
- (b) No amendment to the Plan shall be effective if it authorizes or permits any part of the Plan Fund (other than such part as is required to pay taxes and administration expenses) to be used for or diverted to any purpose other than for the exclusive benefit of the Participants or their Beneficiaries or estates; or causes any reduction in the amount credited to the account of the Participant; or causes or permits any portion of the Plan Fund to revert to or become property of any Participating Employer.
- (c) Except as permitted by the Code, no Plan amendment or transaction having the effect of a Plan amendment (such as merger, plan transfer or similar transaction) shall be effective to the extent it eliminates or reduces any "Section 411(d)(6) protected benefits" the result of which is a further restriction on such benefit unless such protected benefits are preserved with respect to benefits accrued as of the later of the adoption date or effective date of the amendment. "Section

411(d)(6) protected benefits" are benefits described in Code Section 411(d)(6)(A), early retirement benefits and retirement-type subsidies, and optional forms of benefit.

# 8.2 TERMINATION

The Sponsor shall have the right at any time to terminate the Plan by delivering to the Plan Administrator and Benefit Administrator a written notice of such termination. A complete discontinuance of Employer Contributions to the Plan shall be deemed to constitute a termination. Upon any termination (full or partial) or complete discontinuance of contributions, all amounts credited to the affected Participant's Accounts shall not thereafter be subject to forfeiture and all unallocated amounts shall be allocated to the accounts of all Participants in accordance with the provisions thereof. Upon such termination of the Plan, the Sponsor, by written notice to the Plan Administrator and Benefit Administrator, may direct either:

- (a) complete distribution of the assets in the Plan Fund to the Participants, in cash or in kind, in a manner consistent with the requirements of Section 6.3; or
- (b) continuation of the frozen Plan and the distribution of benefits at such time and in such manner as though the Plan had not been frozen.

### 8.3 MERGER OR CONSOLIDATION

This Plan may be merged or consolidated with, or its assets and/or liabilities may be transferred to any other plan only if the benefits which would be received by a Participant of this Plan, in the event of a termination of the Plan immediately after such transfer, merger or consolidation, are at least equal to the benefits the Participant would have received if the Plan had terminated immediately before the transfer, merger or consolidation.

#### 9. MISCELLANEOUS

### 9.1 PARTICIPANT'S RIGHTS

This Plan shall not be deemed to constitute a contract between the Participating Employer and any Participant or to be a consideration or an inducement for the employment of any Participant or Employee. Nothing contained in the Plan shall be deemed to give any Participant or Employee the right to be retained in the service of any Participating Employer or to interfere with the right of the Participating Employer to discharge any Participant or Employee at any time regardless of the effect which such discharge shall have upon them as a Participant of this Plan.

### 9.2 ALIENATION

(a) Subject to the exceptions provided below, no benefit which shall be payable out of the Plan Fund to any person (including a Participant or his or her Designated Beneficiary) shall be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, or charge, and any attempt to anticipate, alienate, sell, transfer, assign, pledge, encumber, or

charge the same shall be void; and no such benefit shall in any manner be liable for, or subject to, the debts, contracts, liabilities, engagements, or torts of any such person, nor shall it be subject to attachment or legal process for or against such person, and the same shall not be recognized by the Plan Administrator, except to such extent as may be required by law.

(b) This provision shall not apply to any "qualified domestic relations order" defined in Code Section 414(p), and those other domestic relations orders permitted to be so treated by the Plan Administrator under such Code Section. The Plan Administrator shall establish a written procedure to determine the qualified status of domestic relations orders and to administer distributions under such qualified orders. Further, to the extent provided under a "qualified domestic relations order" a former spouse of a Participant shall be treated as the spouse or surviving spouse for all purposes of the Plan.

### 9.3 CONSTRUCTION OF PLAN

This Plan shall be construed and enforced according to the Code and the laws of the State of Washington, other than its laws respecting choice of law, to the extent not preempted by the Code or other federal laws. In the event that any Section or Sections of the Code or other federal laws are later determined by any court, legislative body or regulatory agency to be inapplicable to the Plan specifically, or to governmental entity plans in general, then the Plan shall be operated and interpreted in light of any such determination.

### 9.4 GENDER AND NUMBER

Wherever any words are used herein in the masculine, feminine or neuter gender, they shall be construed as though they were also used in another gender in all cases where they would so apply, and whenever any words are used herein in the singular or plural form, they shall be construed as though they were also used in the other form in all cases where they would so apply.

### 9.5 RECOVERY OF ERRONEOUS CONTRIBUTIONS

In the event a Participating Employer shall make an excessive contribution under a mistake of fact, the Participating Employer may recover such excessive contribution at any time within one (1) year following the discovery of the error and the Benefit Administrator shall cause the Trustee to return such amount to the Participating Employer within the one (1) year period.

#### 9.6 SPONSOR'S PROTECTIVE CLAUSE

Neither the Sponsor nor its successors, shall be responsible for the validity of any Contract issued hereunder or for the failure on the part of the insurer issuing such Contract to make payments provided by any such Contract, or for the action of any person which may delay payment or render a Contract null and void or unenforceable in whole or in part.

### 9.7 RECEIPT AND RELEASE FOR PAYMENTS

Any payment to any Participant, his or her legal representative, Designated Beneficiary, or to any guardian or committee appointed for such Participant or Designated Beneficiary in accordance with the provision of the Plan, shall, to the extent thereof, be in full satisfaction of all claims hereunder. This shall include any and all claims against the Sponsor, Plan Administrator and the Participating Employer, any of whom may require such Participant, legal representative, Designated Beneficiary, guardian or committee, as a condition precedent to such payment, to execute a receipt and release thereof in such form as shall be determined by the Sponsor.

### 9.8 ACTION BY AN EMPLOYER

Whenever any Participating Employer under the terms of the Plan is permitted or required to do or perform any act or matter or thing, it shall be done and performed by a person duly authorized by its legally constituted authority.

#### 9.9 HEADINGS

The headings and subheadings of this Plan have been inserted for convenience of reference and are to be ignored in any construction of the provision hereof.

#### 9.10 UNIFORMITY

All provisions of this Plan shall be interpreted and applied in a uniform, nondiscriminatory manner.

#### 9.11 COMPLIANCE WITH USERRA AND RELATED LAWS

Notwithstanding any provision of the Plan to the contrary, service shall be credited and contributions shall be provided to Eligible Employees with respect to qualified military service to the extent required by the Uniformed Services Employment and Reemployment Rights Act of 1994, as such act may be amended from time to time, and as codified in Code Section 414(u) ("USERRA"). Effective January 1, 2007, if a Participant or Former Participant dies while performing qualified military service (as defined in Code Section 414(u)), the Participant's or Former Participant's spouse or other Beneficiary shall be entitled to any benefits (including Employer Contributions relating to the period of qualified military service) and the rights and features associated with those benefits which would have been provided if the Participant or Former Participant had been reemployed by a Participating Employer and died while an active Eligible Employee.

## 9.12 PROHIBITION AGAINST DIVERSION OF FUNDS

(a) Except as provided below and otherwise specifically permitted by law, it shall be impossible by operation of the Plan or the Plan Fund, by termination of either, by power of revocation or amendment, by the happening of any contingency, by collateral arrangement or by any other means, for any part of the corpus or income of any Plan Fund maintained pursuant to

the Plan or any funds contributed thereto to be used for, or diverted to, purposes other than the exclusive benefit of Participants, Retired Participants, or their Beneficiaries.

(b) In the event the Participating Employer shall make an excessive contribution under a mistake of fact, the Participating Employer may demand repayment of such excessive contribution at any time within one (1) year following the time of payment and the Trustees shall return such amount to the Participating Employer within the one (1) year period. Earnings of the Plan attributable to the excess contributions may not be returned to the Participating Employer but any losses attributable thereto must reduce the amount so returned.

### 10. PARTICIPATING EMPLOYERS

### 10.1 REQUIREMENTS OF PARTICIPATING EMPLOYERS

- (a) Each Participating Employer shall be deemed, as a condition of its Participating Employer status, to acknowledge and accept the Plan Administrator, Investment Advisory Committee, Trustee, Benefit Administrator and any other fiduciaries or service providers appointed by the Sponsor or Plan Administrator.
- (b) The Plan Administrator may, but shall not be required to, direct the Trustee to commingle, hold and invest as one Plan Fund all contributions made by Participating Employers, as well as all increments thereof.
- (c) Any expenses of the Trust which are to be paid by the Participating Employer or borne by the Plan Fund shall be paid by each Participating Employer in the same proportion that the total amount standing to the credit of all Participants employed by such Participating Employer bears to the total standing to the credit of all Participants.

### 10.2 DESIGNATION OF AGENT

Each Participating Employer shall be deemed to be a part of this Plan; provided, however, that with respect to all of its relations with the Plan, each Participating Employer shall be deemed to have designated irrevocably the Sponsor as its agent. Unless the context of the Plan clearly indicates the contrary, the words "Employer" or "Participating Employer" shall be deemed to include each Participating Employer as related to its adoption of the Plan.

# 10.3 PARTICIPATING EMPLOYER'S CONTRIBUTION

All contributions made by a Participating Employer, as provided for in this Plan, shall be determined separately by each Participating Employer, and shall be paid to and held by the Plan for the exclusive benefit of the Participants who are Employees or former Employees of such Participating Employer (and the Beneficiaries of such Employees), subject to all the terms and conditions of this Plan.

### 10.4 AMENDMENT

Amendment of this Plan shall be by the written action of the Sponsor, regardless of whether there is a Participating Employer hereunder at such time. Each Participating Employer, by adopting this Plan, consents to the authority of the Sponsor to amend such Plan.

#### 10.5 PLAN ADMINISTRATOR'S AUTHORITY

The Plan Administrator shall have authority as delegated by the Sponsor and authority to make any and all necessary rules or regulations, binding upon all Participating Employers and all Participants, to effectuate the purposes of this Plan.

## 11. MINIMUM DISTRIBUTION REQUIREMENTS

### 11.1 GENERAL RULES

- (a) <u>Effective Date</u>. The provisions of this Article 11 will apply for purposes of determining required minimum distributions for calendar years beginning with the 2002 calendar year that are made on or after January 1, 2002.
- (b) <u>Coordination with Minimum Distribution Requirements Previously in Effect</u>. Required minimum distributions for 2002 under this Article will be determined as follows. If the total amount of 2002 required minimum distributions under the Plan made to the distributee prior to the effective date of this Article 11 equals or exceeds the required minimum distributions determined under this Article, then no additional distributions will be required to be made for 2002 on or after that date. If the total amount of 2002 required minimum distributions under the Plan made to the distributee prior to the effective date of this Article is less than the amount determined under this Article, then required minimum distributions for 2002 on and after that date will be determined so that the total amount of required minimum distributions for 2002 made to the distributee will be the amount determined under this Article.
- (c) <u>Precedence</u>. The requirements of this Article 11 will take precedence over any inconsistent provisions of the Plan.
- (d) <u>Requirements of Regulations Incorporated</u>. All distributions required under this Article 11 will be determined and made in accordance with the Regulations under Code Section 401(a)(9).

### 11.2 TIME AND MANNER OF DISTRIBUTION

(a) <u>Required Beginning Date</u>. The Participant's entire interest will be distributed, or begin to be distributed, to the Participant no later than the Participant's "required beginning date."

- (b) <u>Death of Participant Before Distributions Begin</u>. If the Participant dies before distributions begin, the Participant's entire interest will be distributed, or begin to be distributed, no later than as follows:
  - (1) If the Participant has a Designated Beneficiary, then except as provided in Section 11.2(d) below, distributions to the Designated Beneficiary will begin by December 31 of the calendar year immediately following the calendar year in which the Participant died. Notwithstanding the preceding sentence, if the Participant's surviving spouse is the Participant's sole Designated Beneficiary, distribution is not required to begin until the December 31 of the calendar year in which the Participant would have attained age 70½.
  - (2) If there is no Designated Beneficiary as of September 30 of the year following the year of the Participant's death, the Participant's entire interest will be distributed in the form of a single lump sum distribution by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
  - (3) If the Participant's surviving spouse is the Participant's sole Designated Beneficiary and the surviving spouse dies after the Participant but before distributions to the surviving spouse begin, this Section 11.2(b) (other than Section 11.2(b)(1) as it applies to surviving spouses) will apply as if the surviving spouse were the Participant.

For purposes of this Section 11.2(b) and Section 11.4, unless Section 11.2(b)(3) applies, distributions are considered to begin on the Participant's "required beginning date." If Section 11.2(b)(3) applies, distributions are considered to begin on the date distributions are required to begin to the surviving spouse under Section 11.2(b)(1). If distributions under an annuity purchased from an insurance company irrevocably commence to the Participant before the Participant's "required beginning date" (or to the Participant's surviving spouse before the date distributions are required to begin to the surviving spouse under 11.2(b)(1)), the date distributions are considered to begin is the date distributions actually commence.

- (c) <u>Forms of Distribution</u>. Unless the Participant's interest is distributed in the form of an annuity purchased from an insurance company or in a single sum on or before the "required beginning date," as of the first "distribution calendar year" distributions will be made in accordance with Sections 11.3 and 11.4. If the Participant's interest is distributed in the form of an annuity purchased from an insurance company, distributions thereunder will be made in accordance with the requirements of Code Section 401(a)(9) and the Regulations.
- (d) <u>Distribution Election</u>. Notwithstanding Section 11.2(b), Participants or Designated Beneficiaries may elect on an individual basis whether the five-year rule or the life expectancy rule pursuant to Section 11.2(b) and Section 11.4(c) applies to distributions after the death of a Participant who has a Designated Beneficiary. The election must be made no later than the earlier of September 30 of the calendar year in which distribution would be required to begin or by September 30 of the calendar year which contains the fifth anniversary of the participant's (or if applicable, the surviving spouse's) death. If neither the Participant nor beneficiary makes an

election under this paragraph, distributions will be made in accordance with Section 11.2(b) and Section 11.4(c).

## 11.3 REQUIRED MINIMUM DISTRIBUTIONS

- (a) <u>Amount of Required Minimum Distribution for Each Distribution Calendar Year</u>. During the Participant's lifetime, the minimum amount that will be distributed for each "distribution calendar year" is the lesser of:
  - (1) the quotient obtained by dividing the "Participant's account balance" by the distribution period in the Uniform Lifetime Table in Regulation Section 1.401(a)(9)-9, using the Participant's age as of the Participant's birthday in the "distribution calendar year;" or
  - (2) if the Participant's sole Designated Beneficiary for the "distribution calendar year" is the Participant's spouse, the quotient obtained by dividing the "Participant's account balance" by the number in the Joint and Last Survivor Table in Regulation Section 1.401(a)(9)-9, using the Participant's and spouse's attained ages as of the Participant's and spouse's birthdays in the "distribution calendar year."
- (b) <u>Lifetime Required Minimum Distributions Continue Through Year of Participant's Death.</u> Required minimum distributions will be determined under this Section 11.3 beginning with the first "distribution calendar year" and up to and including the "distribution calendar year" that includes the Participant's date of death.

# 11.4 REQUIRED MINIMUM DISTRIBUTIONS AFTER DEATH

- (a) <u>Death On or After Date Distributions Begin and Participant Survived by Designated Beneficiary</u>. If the Participant dies on or after the date distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each "distribution calendar year" after the year of the Participant's death is the quotient obtained by dividing the "Participant's account balance" by the longer of the remaining "life expectancy" of the Participant or the remaining "life expectancy" of the Participant's Designated Beneficiary, determined as follows:
  - (1) The Participant's remaining "life expectancy" is calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.
  - (2) If the Participant's surviving spouse is the Participant's sole Designated Beneficiary, the remaining "life expectancy" of the surviving spouse is calculated for each "distribution calendar year" after the year of the Participant's death using the surviving spouse's age as of the spouse's birthday in that year. For "distribution calendar years" after the year of the surviving spouse's death, the remaining "life expectancy" of the surviving spouse is calculated using the age of the surviving spouse as of the spouse's birthday in the calendar year of the spouse's death, reduced by one for each subsequent calendar year.

- (3) If the Participant's surviving spouse is not the Participant's sole Designated Beneficiary, the Designated Beneficiary's remaining "life expectancy" is calculated using the age of the beneficiary in the year following the year of the Participant's death, reduced by one for each subsequent year.
- (b) <u>Death On or After Date Distributions Begin and No Designated Beneficiary</u>. If the Participant dies on or after the date distributions begin and there is no Designated Beneficiary as of September 30 of the year after the year of the Participant's death, the minimum amount that will be distributed for each "distribution calendar year" after the year of the Participant's death is the quotient obtained by dividing the "Participant's account balance" by the Participant's remaining "life expectancy" calculated using the age of the Participant in the year of death, reduced by one for each subsequent year.
- (c) <u>Death Before Date Distributions Begin</u>. If the Participant dies before distribution begin, distributions shall be determined as follows:
  - (1) Participant Survived by Designated Beneficiary. Except as provided in Section 11.2(d), if the Participant dies before the date distributions begin and there is a Designated Beneficiary, the minimum amount that will be distributed for each "distribution calendar year" after the year of the Participant's death is the quotient obtained by dividing the Participant's Account balance by the remaining "life expectancy" of the Participant's Designated Beneficiary, determined as provided in Section 11.4(a).
  - (2) <u>No Designated Beneficiary</u>. If the Participant dies before the date distributions begin and there is no Designated Beneficiary as of September 30 of the year following the year of the Participant's death, distribution of the Participant's entire interest will be completed by December 31 of the calendar year containing the fifth anniversary of the Participant's death.
  - (3) Death of Surviving Spouse Before Distributions to Surviving Spouse Are Required to Begin. If the Participant dies before the date distributions begin, the Participant's surviving spouse is the Participant's sole Designated Beneficiary, and the surviving spouse dies before distributions are required to begin to the surviving spouse under Section 11.2(b)(1), this Section 11.4(c) will apply as if the surviving spouse were the Participant.

### 11.5 DEFINITIONS

- (a) The Designated Beneficiary is the individual who is designated as the Beneficiary under Section 1.4 and for purposes of Code Section 401(a)(9) and Regulation Section 1.401(a)(9)-1, Q&A-4.
- (b) The "distribution calendar year" is a calendar year for which a minimum distribution is required. For distributions beginning before the Participant's death, the first distribution calendar year is the calendar year immediately preceding the calendar year that contains the Participant's

"required beginning date." For distributions beginning after the Participant's death, the first distribution calendar year is the calendar year in which distributions are required to begin under Section 11.2(b). The required minimum distribution for the Participant's first distribution calendar year will be made on or before the Participant's "required beginning date." The required minimum distribution for other distribution calendar years, including the required minimum distribution for the distribution calendar year in which the Participant's "required beginning date" occurs, will be made on or before December 31 of that distribution calendar year.

- (c) "Life expectancy" is the life expectancy as computed by use of the Single Life Table in Regulation Section 1.401(a)(9)-9.
- (d) "Participant's account balance" is the Participant's Account balance as of the last Valuation Date in the calendar year immediately preceding the "distribution calendar year" (valuation calendar year), increased by the amount of any contributions made and allocated or forfeitures allocated to the account balance as of dates in the valuation calendar year after the Valuation Date, and decreased by distributions made in the valuation calendar year after the Valuation Date. The account balance for the valuation calendar year includes any amounts rolled over or transferred to the Plan either in the valuation calendar year or in the "distribution calendar year" if distributed or transferred in the valuation calendar year.
- (e) "Required beginning date" is the later of (1) the April 1 of the calendar year following the calendar year in which the Participant attains age 70 ½ or (2) the April 1 of the calendar year following the calendar year in which the Participant retires.

IN WITNESS WHEREOF, this Plan has been executed this 29<sup>th</sup> day of October, 2015, to be effective as of January 1, 2016.

WASHINGTON STATE BOARD
FOR COMMUNITY AND TECHNICAL COLLEGES

Ву	

## **Appendix A: List of Participating Employers**

The Participating Employers in this Plan, as of January 1, 2016, are as follows:

Bates Technical College

Bellevue College

Bellingham Technical College

Big Bend Community College

Cascadia College

Centralia College

Clark College

Clover Park Technical College

Columbia Basin College

Community Colleges of Spokane (District 17)

**Edmonds Community College** 

**Everett Community College** 

Grays Harbor College

Green River College

Highline College

Lake Washington Technical College

Lower Columbia College

Olympic College

Peninsula College

Pierce College (District 11)

Renton Technical College

Seattle Colleges (District 6)

Shoreline Community College

Skagit Valley College

South Puget Sound Community College

Tacoma Community College

Walla Walla Community College

Washington State Board for Community and Technical Colleges

Washington Student Achievement Council

Wenatchee Valley College

Whatcom Community College

Yakima Valley Community College

